Author's response to reviews

Title: A rare case of isolated thyroid metastasis revealing an unknown lung adenocarcinoma

Authors:

Jihane KHALIL (jihane.khalil@gmail.com)
FADOUA ELOMRANI (elfadwa512@hotmail.fr)
MERYEM BENOULAID (mery-ben23@hotmail.com)

Version: 5  Date: 15 July 2015

Author's response to reviews: see over
Dear Reviewers,

I hope my email finds you well. I would like to sincerely thank you for your kind words, dedicating some of your precious time to review my manuscript and offering valuable suggestions.

To reviewer n.2:

*Had the patient neck pain or goiter enlargement, or weight loss or other not specific symptoms of malignancy?*

In fact, the patient did not present any of these symptoms. The diagnosis of the thyroid mass was made during routine follow up.

*There is a mistake in line 4 of the discussion: it is has been*

I have changed it to It has been. Thank you